Author's response to reviews

Title: Obliteration of radical cavities with autogenous cortical bone; long-term results

Authors:

Akram M Abdel-Rahman (akram.abdelrahman@gmail.com)
Matti Pietola (matti.pietola@hus.fi)
Teemu J Kinnari (teemu.j.kinnari@helsinki.fi)
Hans Ramsay (hans.ramsay@hus.fi)
Jussi Jero (jussi.jero@hus.fi)
Antti A Aarnisalo (antti.aarnisalo@helsinki.fi)

Version: 6  Date: 15 July 2008

Author's response to reviews:

Dear Editor,

-The tables have been revised according to your instructions.
-The questionnaire is added as an appendix to the manuscript

Kind regards,
Antti Aarnisalo MD PhD